Author's response to reviews

Title: Combined determination of plasma MMP2, MMP9, and TIMP1 improves the non-invasive detection of transitional cell carcinoma of the bladder

Authors:

Andrea Staack (andrea.staack@charite.de)
Steffen Badendeck (badendeck@12move.de)
Dietmar Schnorr (dietmar.schnorr@charite.de)
Stefan Loening (stefan.loening@charite.de)
Klaus Jung (klaus.jung@charite.de)

Version: 7 Date: 24 July 2006

Author's response to reviews: see over
RE: Resubmission of original work manuscript:
No. 1736728878568775

Combined determination of plasma MMP2, MMP9, and TIMP1 improves the non-invasive detection of transitional cell carcinoma of the bladder

Andrea Staack, Steffen Badendieck, Dietmar Schnorr, Stefan A. Loening, Klaus Jung

From the Department of Urology, University Hospital Charité, Humboldt University of Berlin, Germany

Dear Members of the Editorial Board,

We are glad to hear about the acceptance of our manuscript.

The following changes have been made in the text as requested:

  p. 6, line 7-8  From each patient a verbal informed consent was obtained.

We also went through the format guidelines and applied some small format changes.

Yours sincerely,

Andrea Staack